Author's response to reviews

Title: Musculoskeletal symptoms of the upper extremities and the neck: a cross-sectional study on prevalence and symptom-predicting factors at visual display terminal (VDT) workstations

Authors:

Andre Klussmann (klussmann@uni-wuppertal.de)
Hansjuergen Gebhardt (aser1@uni-wuppertal.de)
Falk Liebers (liebers.falk@baua.bund.de)
Monika A Rieger (monika.rieger@uni-wh.de)

Version: 4 Date: 26 June 2008

Author's response to reviews:

Dear BioMed Central Editorial Production Team,

Thank you for your rapid response.

We performed all formatting changes requested:

- removed the words "Authors, Address, Email" from the title page
- presented the author list as a sentence
- removed the keywords
- refrained from using underlining, italics, and bold fonts in the text - except for section headings
- re-titled "Figures" in "Figure legends".

Once again we would like to thank the BioMed Central Editorial and Editorial Production Team for the very competent and friendly management of the publication process.

We are looking forward to the publication of the manuscript.

Yours sincerely,
Monika A. Rieger, M.D.